Author's response to reviews

Title: The NTI-tss device for the therapy of bruxism, temporomandibular disorders, and headache: where do we stand? A qualitative systematic review of the literature

Authors:

Henrike Stapelmann (h.stapelmann@unibas.ch)
Jens C Turp (jens.tuerp@unibas.ch)

Version: 6 Date: 25 June 2008

Author's response to reviews: see over
Dear Editorial Team,

Following your request from 23 June 2008, we have carried out the following formatting changes:

1. **Methods:** We have removed the website address <www.fda.gov/search.html> from the text. Since in the previous version of our manuscript we had removed all web addresses of the databases without including citations to these addresses in the reference list, we desisted to do so for the FDA Medical Device website for the sake of uniformity. Instead, the sentence in question now reads:
   “In addition, a specific search in the FDA’s Manufacturer and User Facility Device Experience Database (MAUDE) was carried out with the following strategy: Go to Simple Search → Search term: NTI; Date Report Received by FDA: ALL YEARS”.

2. We have included the former Table 13 as Figure 3 (PDF). As a consequence, we have renumbered former Figure 3 (now Figure 4) as well as former Tables 14 to 21 (now Tables 13 to 20).

3. We would like to accept your kind offer to re-orientate Tables 14 to 18 for us.

4. Due to the problems mentioned in our letter from 19 June 2008, we have submitted the manuscript (which has been generated by using EndNote) in Rich Text Format.

We hope that we have executed all point correctly now and would like to thank you for the support that you gave us to generate an acceptable manuscript.

Sincerely,

Professor